Author's response to reviews

Title: Female genital mutilation of a karyotypic male presenting as a female with delayed puberty

Authors:

Mona Ellaithi (Ellaithi_mona@yahoo.com)
Therese Nilsson (Therese.Nilsson@med.lu.se)
David Gisselsson (David.Gisselsson@med.lu.se)
Atif Elagib (Atifelaqib@yahoo.com)
Hamid Eltigani (Heeltigani@yahoo.com)
Imad Fadi-Elmula (Fadi-elmula@neelain.info)

Version: 7 Date: 14 March 2006

Author's response to reviews:

Dear Editor,
Thank you for accepting our manuscript for publication in your journal; it is a great honor to us. Please find enclosed the revised version of our manuscript Female Genital Mutilation of a karyotypic male presenting as a case of female with delayed puberty.
The manuscript has been revised by all co-authors for final spelling and other minor changes you requested. No changes were done to the final formatting of the manuscript that you have already been accepted.

Thank you for your care,
Mona Ellaithi
Corresponding author